Author's response to reviews

Title: Physiological and neurophysiological determinants of postcancer fatigue: design of a randomized controlled trial

Authors:

Hetty Prinsen (h.prinsen@onco.umcn.nl)
Gijs Bleijenberg (g.bleijenberg@nkcv.umcn.nl)
Machiel J. Zwarts (zwartsm@kempenhaeghe.nl)
Maria T.E. Hopman (m.hopman@fysiol.umcn.nl)
Arend Heerschap (a.heerschap@rad.umcn.nl)
Hanneke W.M. van Laarhoven (h.vanlaarhoven@amc.uva.nl)

Version: 2 Date: 5 June 2012

Author's response to reviews: see over
Dear editor in chief,

Please find enclosed our manuscript entitled “Physiological and neurophysiological determinants of postcancer fatigue: design of a randomized controlled trial”, which we would like to resubmit for publication as study protocol in BioMed Central Cancer.

We have made no changes to the manuscript or to the accompanying figure itself, but we have sent you by email the documents and the information you asked for, including:

1. Our study was ethical approved by the local ethics committee of the Radboud University Nijmegen Medical Centre. I have sent you a copy of this approval.

2. Our study was externally funded by the Dutch Cancer Society. I have sent you a copy of the proof of external funding.

3. The first results of this study are under submission with Supportive Care in Cancer.

Looking forward to your response,

On behalf of all the co-authors,
Sincerely yours,

Hetty Prinsen.

Authors: Hetty Prinsen, Gijs Bleijenberg, Machiel Zwarts, Maria Hopman, Arend Heerschap, and Hanneke van Laarhoven.

Corresponding author’s contact information: M.Sc. Hetty Prinsen, Department of Medical Oncology, Radboud University Nijmegen Medical Centre, P.O. Box 9101, 6500 HB Nijmegen, the Netherlands. Phone: +31 24 3610354; Fax: +31 24 3540788; E-mail: h.prinsen@onco.umcn.nl